Author’s response to reviews

Title: Unresponsive wakefulness syndrome: a new name for the vegetative state or apallic syndrome?

Authors:

Steven Laureys (steven.laureys@ulg.ac.be)
Gastone Celesia (g.celesia@comcast.net)
Francois Cohadon (francois.cohadon@neuf.fr)
Jan Lavrijsen (j.lavrijsen@vphg.umcn.nl)
José Léon-Carrrión (leon carrrión@us.es)
Walter G Sannita (wgs@dism.unige.it)
Leon Sazbon (sazbon@zahav.net.il)
Erich Schmutzhard (erich.schmutzhard@i-med.ac.at)
Klaus R von Wild (kvw@neurosci.de)
Adam Zeman (adam.zeman@pms.ac.uk)
Giuliano Dolce (g.dolce@istitutosantanna.it)
European Task force on Disorders of Consciousness - (coma@chu.ulg.ac.be)

Version: 6 Date: 8 October 2010

Author’s response to reviews:

Formatting changes to be made:
---------------------------------

Major revisions (we require the author to make these changes)

Author list: if you wish to include the European Task force on Disorders of
Consciousness on the title page, please add it to the author list, assign it an
author affiliation and add it to the submissions system with an email address.
Alternatively, please remove the group from the title page and mention in the
Acknowledgements section that the authors submitted this manuscript on behalf
of the group.

We have included the European Task force on Disorders of Consciousness on
title page and added to author list - same affiliation as first author - email
coma@chu.ulg.ac.be

Steven Laureys